

Edinburgh Research Explorer

Camouflaging, not sensory processing or Autistic identity, predicts eating disorder symptoms in Autistic adults

Citation for published version:

Bradley, S, Moore, F, Duffy, F, Clark, L, Suratwala, T, Knightsmith, P & Gillespie-Smith, K 2024, 'Camouflaging, not sensory processing or Autistic identity, predicts eating disorder symptoms in Autistic adults', *Autism*, pp. 1-11. https://doi.org/10.1177/13623613241245749

Digital Object Identifier (DOI): 10.1177/13623613241245749

Link:

Link to publication record in Edinburgh Research Explorer

Document Version:

Publisher's PDF, also known as Version of record

Published In:

Autism

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

The University of Edinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





Original Article



Camouflaging, not sensory processing or autistic identity, predicts eating disorder symptoms in autistic adults

Autism
I-II
© The Author(s) 2024



Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/13623613241245749 journals.sagepub.com/home/aut



Siofra Bradley¹, Fhionna Moore², Fiona Duffy^{1,3}, Lili Clark¹, Tasha Suratwala⁴, Pooky Knightsmith⁵ and Karri Gillespie-Smith¹

Abstract

The objective of this study was to explore the role that Autistic identity, sensory processing and camouflaging behaviours have on eating disorder symptomology in Autistic adults. Previous research has focused on cognitive and sensory factors to explain the association between autism and eating disorders, but the roles of social identity and camouflaging are yet to be explored. Autistic participants (N=180) were recruited from NHS settings and community groups. The participants completed online questionnaires measuring autistic identity, camouflaging behaviours, sensory processing, autistic traits and eating disorder symptoms. Multiple regression revealed that camouflaging significantly predicted eating disorder symptoms. Although sensory processing was related, it did not significantly *predict* eating disorder symptom severity. In addition, there was no significant relationship between autistic identity and eating disorder symptom severity. This study highlights the impact that camouflaging behaviours and sensory processing can have on eating disorder symptomatology in autism and may indicate important considerations for the treatment of eating disorders in Autistic people.

Lay Abstract

This study aimed to explore the impact of Autistic identity (i.e. feeling like you belong to the Autistic community), sensory profiles (e.g. being over or under responsive to sensations) and camouflaging behaviours (i.e. masking) on eating disorder symptoms in Autistic adults. 180 Autistic people were recruited from the community and NHS. The Autistic people completed online questionnaires measuring Autistic identity, sensory profiles, camouflaging behaviours, autistic traits and eating disorder symptoms. The analysis showed that higher levels of camouflaging behaviour predicted higher levels of eating disorder symptoms. Sensory profiles were related to but did not predict eating disorder symptoms and there was no relationship between level of Autistic identity and eating disorder symptoms. This shows that camouflaging is the most important predictor of eating disorder symptoms in Autistic people, and warrants further exploration.

Keywords

autism, Autistic identity, camouflaging, eating disorders, sensory processing

Introduction

Autism and eating disorders

Autism¹ is a lifelong neurodevelopmental condition characterised by differences in communication, social functioning, patterns of behaviour and restricted interests (American Psychiatric Association (APA), 2013). A recent systematic review reported the global prevalence of autism to be around 1% of the population, with a male to female ratio of 4:2 (Zeidan et al., 2022). However, these figures must be interpreted with caution, given the issues associated with autism assessment cost, as well as the underdiagnosis of Autistic females (Adamson et al., 2020; Hull & Mandy, 2017; Milner et al., 2019).

Research has consistently shown that Autistic people have higher rates of other neurodivergent conditions and mental health difficulties than non-Autistic people, including eating disorders (EDs) (Lai et al., 2019). EDs include

¹The University of Edinburgh, UK

²NHS Tayside, UK

³NHS Lothian, UK

⁴NHS England, UK

⁵Creative Education, UK

Corresponding author:

Karri Gillespie-Smith, Clinical and Health Psychology, School of Health in Social Science, Old Medical School, The University of Edinburgh, Teviot Place, Edinburgh EH8 9AG, UK. Email: karri.gillespie-smith@ed.ac.uk

anorexia nervosa (AN), bulimia nervosa (BN), binge eating disorder (BED) and, more recently, avoidant/restrictive food intake disorder (ARFID; which is a feeding disorder; Carter Leno et al., 2022). NICE (2017) reports an overall estimate of 700,000 people in the United Kingdom with a diagnosis of EDs, 90% of whom are females. However, it is recognised that EDs in males are significantly underdiagnosed and only 1% of ED research is conducted with males (Lavender et al., 2017; Murray et al., 2017).

Despite limited prevalence research, reports show that Autistic individuals are twice as likely to experience EDs compared with their neurotypical counterparts (Sedgewick et al., 2021).

Systematic reviews have reported autism prevalence rates within clinical populations of AN and BN to range between 4.7% and 22.9% (Huke et al., 2013; Nickel et al., 2019). Positive relationships between autistic traits and ED symptoms have also been found in non-clinical populations (Barnett et al., 2021; Christensen et al., 2019).

Those who are Autistic and have EDs have been found to experience longer admissions in inpatient settings, with more severe presentations of EDs and poorer outcomes (Nielsen et al., 2015, 2022). Autistic people who have had ED treatment have reported their treatment experiences as being characterised by a lack of understanding and accommodations for autism, which has impacted their recovery and relationships with clinicians (Kinnaird et al., 2019). Due to a lack of research, there is no robust clinical guidance on treatment options for this population which may explain why ED clinicians have shared that they feel underconfident treating people with this presentation (Kinnaird et al., 2017).

As far as the authors are aware, to date, there has only been one clinical pathway designed for treating Autistic people with AN, taking into account sensory and cognitive profiles common to Autistic people. This is the PEACE (Pathway for Eating Disorders and Autism developed from Clinical Experience) Pathway (Tchanturia et al., 2021). This has showed extremely promising results, including reduced length of inpatient stay, increased clinician confidence in working with Autistic people with AN, and NHS savings equating to approximately £22,837 per patient and approximately £275,000 per year for the ED service as a whole (Tchanturia et al., 2021).

Explanations for ED in autism

Up until now, both quantitative and qualitative research have focused on cognitive, social and sensory aspects of autism that may contribute towards the development and maintenance of EDs among Autistic people (Brede et al., 2020; Christensen et al., 2019; Kinnaird et al., 2019). Significant levels of cognitive factors associated with autism such as inflexible thinking, rigidity and theory of mind difficulties have been reported among ED populations, such as those with AN (Christensen et al., 2019; Kinnaird et al., 2019), but there is less evidence of this

association in other EDs such as BED (Kelly et al., 2013; Manasse et al., 2015).

The directionality of this association remains a topic of debate. Seemingly, Autistic characteristics such as cognitive rigidity may be prevalent in those with EDs as a temporary result of starvation on the brain (Hiller & Pellicano, 2013). In contrast, some evidence has shown that in women who have recovered from EDs, high levels of autistic characteristics have still been found (Bentz et al., 2017; Nazar et al., 2018). Furthermore, there is growing evidence to suggest that autistic traits from childhood such as social communication difficulties may increase vulnerability for developing EDs (Westwood et al., 2018). Solmi et al.'s (2021) longitudinal research demonstrated that autistic traits in 7-year-olds predicted their disordered eating behaviour at age 14.

EDs and sensory processing. The relationship between sensory processing issues and common eating behaviours in autism has recently been highlighted by Nimbley and colleagues' (2022) systematic review. Different sensory profiles such as hyper- and hyposensitivities have been found among Autistic people (Crane et al., 2009) and it has been proposed that these sensitivities can contribute to EDs. For example, food restriction may be a coping mechanism for distressing hyper-sensory experiences (Brand-Gothelf, 2016; Kinnaird et al., 2020), while hyposensitivity to internal sensations may result in difficulties interpreting hunger cues (Brede et al., 2020).

Brede et al.'s (2020) qualitative study with parents, healthcare workers and Autistic women with AN extracted themes of sensory overload, food-specific sensory sensitivities and internal and bodily sensations as contributing to Autistic women's EDs. Treatment settings such as those with bright or loud inpatient wards and meal plans involving certain food textures, smells and tastes can make ED recovery particularly challenging (Kinnaird et al., 2019).

EDs and social identity. A theme highlighted by qualitative accounts and case studies of Autistic people with EDs is the role that social identity played in the development and maintenance of their ED (Brede et al., 2020; Dandil et al., 2019). It has been proposed that the development of an ED could be related to identity issues that arise from being Autistic. Autistic people are part of a minority and are often a stigmatised group both before and after diagnosis (Botha & Frost, 2020; MacLeod et al., 2013). Accounts of Autistic people with EDs have suggested that AN may provide an identity or group to belong to (i.e. people who have AN) or be a form of coping due to not fitting in (Brede et al., 2020). This taken together highlights the need to investigate the role that Autistic identity plays on ED symptomology.

Tajfel et al.'s (1979) Social Identity Theory (SIT) framework suggests that if people are a member of a stigmatised group, they engage in attempts to regain a positive identity. They do this by either promoting the in-group status (joining support groups or Autism rights organisations) or dissociating from this group and identifying with a more

'socially accepted' group. Many Autistic people are rightly proud of their Autistic identity and appreciate the strengths that are associated with being Autistic (R. Cooper et al., 2021). Positive social identification with Autism has been associated with mental health benefits such as better self-esteem and lower levels of anxiety and depression (Cage et al., 2018; K. Cooper et al., 2017; R. Cooper et al., 2021). However, this has not been explored as a potential protective factor against EDs specifically in Autistic people.

EDs and camouflaging. 'Camouflaging' or 'masking' is an attempt to dissociate from autism by engaging in behaviours that appear non-Autistic such as maintaining eye contact and rehearsing facial expressions (Bargiela et al., 2016; Beck et al., 2020). Autistic people have reported that camouflaging has been utilised as a response to autism stigma, in an attempt to mask their difficulties, fit in to a non-autistic world and identify with social groups (Bargiela et al., 2016; Hull & Mandy, 2017; Perry et al., 2022). This camouflaging, however, leads to increased levels of mental and physical exhaustion (Bargiela et al., 2016; Beck et al., 2020), with a recent review reporting that higher levels of camouflaging relate to poorer mental health outcomes (Cook et al., 2021). However, the relationship between camouflaging and ED outcomes has not yet been explored.

Similar to ED prevalence, camouflaging appears to be more prevalent among Autistic females than males (Lai et al., 2017) which may evoke speculation that EDs and camouflaging could be interlinked. This has been corroborated by Autistic females' accounts in ED inpatient settings where they often report to engage in camouflaging behaviours (e.g. copying others and adopting their anorexic values) as a way of fitting in to a neurotypical world (Brede et al., 2020). Further evidence has been provided by caregivers who highlight that AN may emerge from a reduced ability to cope in more complex social situations as Autistic young people get older (Adamson et al., 2020). Given the implications of the associations between camouflaging and ED development and maintenance for Autistic people, it is surprising that this relationship has not been investigated.

Study aims

The aim of the current study is to explore the role of Autistic identity, camouflaging behaviours and sensory processing in ED symptomology severity. The purpose of this is to add greater understanding towards the development and maintenance of EDs in autism, providing information to help inform treatment pathways for Autistic people with EDs and prevent poor outcomes.

Study predictions

It is predicted that a strong Autistic identity will predict lower levels of ED symptomology. It is also predicted that higher levels of camouflaging will predict higher levels of ED symptomology, and finally high levels of sensory processing issues will predict high levels of ED symptomology.

Methods

Recruitment

This study received ethical approval from the NHS West Midlands Solihull Research Ethics Committee and NHS Tayside to act as a Participant Identification Centre.

Participants completed questionnaires online via the Qualtrics Survey platform from May 2022 to October 2022. Online recruitment was chosen for this study due to possible disruptions that COVID-19 imposed and because previous research highlighted that Autistic people often prefer online platforms as it reduces social demands (K. Cooper et al., 2017; Hendrickx, 2015). There were no incentives or reimbursements offered for participation in this study.

Clinicians within NHS Tayside Eating Disorder service (TEDS) and Autism service (TAACT) informed eligible patients about the study. Social Media ads containing the link to the online Qualtrics platform were placed on social media platforms, as well as university and charity webpages.

Participants

Participants were included in this study if they were able to read English, were over the age of 18 and had either a clinical or self-diagnosis of Autism. Including people who self-identify as Autistic as this is a more inclusive practice endorsed by the Autistic community. In addition, research shows that those who self-identify as Autistic are accurate at recognising autistic traits in themselves and score similarly in autism scales, employment struggles, stigma and quality of life measures (McDonald, 2020; Overton et al., 2023). Participants were not excluded if they had intellectual disabilities and were asked to indicate this on the demographic section of the survey. No one indicated that they had an intellectual disability.

A priori power calculations using a G*Power calculator (Faul et al., 2009) was used to compute an adequate sample size for a regression analysis with five predictor variables. The calculation was based on a statistical power of 0.80 (Cohen, 1988), an alpha level of 0.5 (Green, 1991) with a medium effect size of F^2 =0.15 which calculated that a sample of 92 participants would be required for this study. The current study initially recruited N=265 participants, of which some had to be removed due to missing or incomplete data (see 'Data analyses' section for more information), leaving the final sample as N=180.

Participants remained anonymous throughout; however, demographic information was gathered. The age range of participants was 18–70 with an average age of 38

(SD=13.6) years. Most participants had a clinical diagnosis (80.6%, N=145) rather than a self-diagnosis (19.4%, N=35) of Autism. Participants mostly consisted of females (64.4%, N=116), followed by males (20%, N=36), transgender (10.6%, N=19) and those who preferred not to say (5%, N=9). Most participants were from a White ethnic background (87.2%, N=157), followed by Mixed or Multiple ethnicities (6.7%, N=12), Black or African American or Caribbean (2.8%, N=5), Asian (2.8%, N=5)and Other (0.6%, N=1). Participants were asked if they have ever received a diagnosis of an ED in their lifetime. The majority of participants (63.9%, N=115) responded 'Yes' while a smaller amount reported 'No' (36.1%, N=65). Table 1 presents the types of diagnosis that people reported as well as the frequency of other mental health diagnoses that people currently had.

Measures

The study contained five questionnaires to gather information on the variables of autistic traits, sensory processing, camouflaging and Autistic identity, with another questionnaire measuring the outcome variable – ED symptomology. It took participants approximately 30 min to complete the study.

Autistic traits. The Autism Spectrum Quotient (AQ-10; Allison et al., 2012) is a shortened 10-item version of the 50-item Autism Quotient scale used to indicate levels of autistic traits, with higher scores indicating more autism characteristics. The maximum score for the AQ-10 is 10. The AQ-10 was used in this study for descriptive purposes to indicate participant autistic trait levels, and was not used as a screening instrument to verify clinical status. Critically, it has previously demonstrated sensitivity of 0.88, specificity of 0.91 and positive predictive value of 0.85 and internal consistency of (α >0.85) among male and female samples (Allison et al., 2012). The AQ-10 displayed moderate reliability in this study (a=0.63).

Camouflaging. Levels of camouflaging were measured using the Camouflaging Autistic Traits Questionnaire (CAT-Q; Hull et al., 2019). This questionnaire contains 25 statements which participants respond to using a Likert-type scale to indicate how much they agree or disagree with each statement. The total scores range from 25 to 175, with higher scores indicating higher camouflaging behaviours. The CAT-Q has previously demonstrated good test-retest reliability (r=0.77) and high internal consistency for camouflaging (α =0.94) (Hull et al., 2019). In this study, the CAT-Q displayed robust internal consistency (α =0.82).

Autistic identity. The Social Identification Scale (SIS; Leach et al., 2008) is a 14-item scale which has shown good reliability (α =0.87) across a range of identities. It

Table 1. Participant demographic information.

Characteristics	N (%)	
Age in years (n=8 missing)	38 (SD: 13.6)	
Gender		
Female	116 (64.4)	
Male	36 (20)	
Transgender	19 (10.6)	
Prefer not to say	9 (5)	
Ethnicity		
White	157 (87.2)	
Mixed or multiple ethnicities	12 (6.7)	
Black or African American or Caribbean	5 (2.8)	
Asian	5 (2.8)	
Other	I (0.6)	
Autism diagnosis		
Clinical diagnosis	145 (80.6)	
Self-diagnosis	35 (19.4)	
Eating disorder diagnosis		
No	115 (63.9)	
Yes	65 (36.1)	
Frequency of eating disorder types		
Multiple	23 (35.3)	
Anorexia nervosa	22 (33.8)	
Bulimia nervosa	6 (9.2)	
Binge disorder	5 (7.7)	
Other	4 (6.2)	
OSFED (other specified feeding and eating disorder)	2 (3.1)	
ARFID (avoidant/restrictive food intake disorder)	2 (3.1)	
Did not say	l (1.5)	
Other mental health diagnosis	` '	
No	64 (36)	
Yes	116 (64.4)	

This table demonstrates the mean age and standard deviation of participants. It also demonstrates the number, total percentage and amount of missing data for each characteristic.

can be adapted to measure Autistic identity by putting the phrase 'Autistic person/people' at the start of each question (K. Cooper et al., 2017) and has shown good internal consistency (α =0.91). Participants rate on a 7-point Likert-type scale how much they agree or disagree with each statement. It has two dimensions: self-investment which is made up of solidarity (α =0.82), satisfaction (α =0.84) and centrality (α =0.78), and self-definition which is comprised of self-stereotyping (α =0.86) and in-group homogeneity (α =0.66). The SIS demonstrated strong internal consistency in this study (α =0.90).

Sensory processing. The Sensory Processing Quotient (SPQ; 10 items; Greenberg et al., 2018) was used to measure participant's levels of sensory sensitivity across the five senses. The SPQ-10 showed good internal consistency from this study (α =0.85). Participants respond to 10

statements using a Likert-type scale to indicate how strongly they agree or disagree with each statement. The maximum score that can be obtained is 30, with higher scores indicating higher levels of sensory sensitivities.

Eating disorder symptom severity. The Eating Disorder Examination Questionnaire Short (EDE-QS; Gideon et al., 2016) was used to measure ED behaviours and attitudes (from here on referred to as ED symptomology) by asking participants to consider the preceding 7 days when responding to 12 items on a 4-point response scale. It is a shortened version of the Eating Disorder Examination Questionnaire (EDE-Q) 28-item commonly used in research concerned with the overlap between autism and EDs (Kerr-Gaffney et al., 2020; Westwood et al., 2017). A maximum score of 36 can be obtained with higher scores indicating more disordered eating symptomology and scores of 15 or above indicating potential ED cases in clinical settings (Prnjak et al., 2020).

The EDE-QS has been highly correlated with the EDE-Q (r=0.91 for people without EDs; r=0.82 for people with EDs) and displayed high internal consistency in this study and in previous studies (α =0.91) (Gideon et al., 2016). It has shown good levels of specificity (0.85), sensitivity (0.83) and positive predictive value (0.37) (Prnjak et al., 2020).

Data analyses

The data were exported from Qualtrics to a password-protected excel file which was only accessible to members of the research team. The data were analysed using IBM SPSS V22. Data cleaning occurred to remove erroneous data before analyses occurred. Data sets were removed if they did not consent to take part (N=32), were less than 85% completed (N=48), completed in less than 260 s as this may have implied automated responses (N=3), and completed as part of a preview or mock responses (N=2). There were no missing data for any of the predictor variables and only 1.7% missing data for the outcome variable (EDE-QS), due to this low proportion no further action was required at the pre-analysis stage.

The data were explored to determine whether it met assumptions for parametric tests. The variance inflation factor (VIF) for each variable was below 2, indicating that the data met the assumption of collinearity. The assumption of normality of residuals yielded conflicting results. Assumptions of normality of residuals were checked through viewing histogram and Q-Q plots, and scatterplots of the standardised residuals and predicted values indicated that the assumptions of homoscedasticity and linearity were also met.

Bivariate correlations were conducted to establish which predictor variables showed a significant relationship with the outcome variable of ED symptomology and

Table 2. Mean score and standard deviation of each variable.

Variable	М	SD	
ED symptoms	6.86	4.86	
AT	7.87	1.90	
SP	19.74	6.37	
СВ	115.48	17.99	
Al	66.32	13.91	

ED symptoms: eating disorder symptoms; AT: Autistic traits; SP: sensory processing; CB: camouflaging behaviours; AI: Autistic identity.

only significant results were taken forward and entered into the regression model. A stepwise multiple regression was then performed to test the study predictions. The autistic traits predictor was entered into the model first as this was the control variable, followed by the predictor variables based on the strength of their relationship with the outcome variable of eating disorder symptom severity. The predictor variable of sensory processing demonstrated the weakest relationship and was entered first, followed by camouflaging. The regression coefficients produced from the multiple regression were explored to examine the unique variance that each predictor variable contributed to the ED symptomology outcome.

Community involvement

This study was co-produced by members from the Autistic community with lived experience following participatory research guidelines (Fletcher-Watson et al., 2019). A peer Autistic researcher with lived experience of EDs helped with the initial conceptualisation of the project, ensuring relevance to the Autistic community. Two Autistic mentors also helped select appropriate measures, looked over participant documents and were involved in the writing (and co-authorship) of this article.

Results

Means and standard deviations

Table 2 presents the means and standard deviations for each of the independent variables (see Table 2).

Correlations

Table 3 presents correlations between the variables that were significant with the outcome variable of disordered eating symptomology (see Table 3).

Pearson's correlation tests were performed with a Bonferroni-adjusted alpha level of 0.013 (0.05/4) to explore the presence and strength of relationships between the variables. Weak to moderate positive and significant correlations were found between ED symptomology and

Table 3. Correlations between variables and ED symptoms.

	AT	Al	СВ	SP	ED symptoms
Autistic traits (AT)	_	0.331** (<0.001)	0.264** (<0.001)	0.230** (<0.013)	0.190* (<0.013)
Autism identity (AI)			0.353** (<0.001)	0.184* (<0.013)	-0.021 (0.761)
Camouflaging behaviours (CB)				0.326** (<0.001)	0.302** (<0.001)
Sensory processing (SP)				_	0.225** (<0.013)

ED: eating disorder.

camouflaging behaviours (r=0.30, p<0.001), sensory processing (r=0.23, p<0.013), and autistic traits (r=0.19, p<0.013). A negative correlation was found between ED symptomology and Autistic identity; however, this was not significant (r=-0.02, p=0.761).

Although it was not related to the study's predictions, interesting relationships were found between variables as shown in Table 3. Camouflaging and Autistic identity showed a significant positive relationship (r=0.35, p<0.001) Camouflaging behaviours showed significant positive relationships with autistic traits (r=0.26, p<0.001), and sensory processing (r=0.33, p<0.001). Sensory processing displayed significant positive relationships with autistic traits (r=0.23, p<0.013).

There was no significant relationship found between ED symptomology and Autistic identity. Therefore, this variable was not included as a predictor variable for ED symptomology in further regression analyses.

Stepwise multiple regression

A stepwise multiple regression was conducted to further analyse the significant relationships found between ED symptomology and the predictor variables of sensory processing and camouflaging. Only three variables were entered into the following regression model which meant the Bonferroni-adjusted alpha applied to the regression analyses was 0.017 (0.5/3). As autistic traits were significant at the p < 0.013 level, they were treated as a control variable and entered into the model first. Model 1, with autistic traits as the only predictor, explained 3.1% of the variance and was significant (F(1, 178) = 6.64, p < 0.017). Model 2, in which Sensory Processing was added, explained significantly more variance (R^2 change=0.035, F(1, 177) = 6.61, p < 0.017). The model significantly predicts 6% of the variance in ED symptomology (adjusted $R^2 = 0.060$) (F(2, 177)=6.73, p<0.017). Model 3, in which camouflaging was added, explained more variance, and this increase was significant (R^2 change=0.047, F(1,176)=9.39, p < 0.017). Model 3 significantly predicted

10.3% of the variance in ED symptomology outcomes (adjusted R^2 =0.103) and was significant (F(3, 176=7.83, p<0.001).

The regression coefficients showed the unique variance that each predictor variable contributed to ED symptomology outcomes. Model 3 shows that camouflaging (β = 0.24, p < 0.016) contributed unique variance to ED symptomology outcomes. Autistic traits (β = 0.10, p = 0.186) and sensory processing (β = 0.13, p = 0.099) did not contribute unique significant variance to ED symptomology outcomes, see Table 4.

Discussion

Findings

This study explored the role that Autistic identity, sensory processing and camouflaging may have on ED symptomology in Autistic people. The predictions were partly supported and the results show that high levels of camouflaging behaviours uniquely predicted ED symptomology in Autistic people. Although sensory processing was related, it was not found to significantly predict ED symptomology. In addition, there was no evidence to support the prediction that strong Autistic identity predicted lower levels of ED symptoms.

Camouflaging and ED symptoms

This is the first study that quantitatively demonstrates the role of camouflaging on ED symptoms in Autistic individuals. These findings corroborate Autistic people and their parent's accounts that EDs are related to camouflaging behaviours and that they are developed as a way of coping with social demands in a neurotypical world (Adamson et al., 2020; Brede et al., 2020). These findings uniquely add to existing research that have reported high levels of camouflaging behaviours are associated with poor mental health outcomes (Bargiela et al., 2016; Beck et al., 2020) by extending this further to ED symptoms.

^{*}Correlation is significant to 0.013 (following Bonferroni correction; two-tailed).

^{**}Correlation is significant to 0.001.

Table 4. Stepwise multiple regression model.

	Variable	В	β	Þ	df	F	Þ	R ² Adj
(EDE-QS)								
Model I					1, 178	6.636	0.001	0.031
	AT	0.486	0.190	0.011				
Model 2					2, 177	6.728	0.001	0.071
	AT	0.373	0.145	0.052				
	SP	0.146	0.191	0.011				
Model 3					3, 176	7.829	0.003	0.118
	AT	0.253	0.099	0.186				
	SP	0.096	0.126	0.099				
	СВ	0.063	0.235	0.003				

AT: Autistic traits; SP: sensory processing; CB: camouflaging behaviours.

Autistic identity and ED symptomology

For the first time, the relationship between Autistic identity and ED symptomology was explored and it is somewhat surprising that a significant relationship was not found between these concepts. Previous research which also measured Autistic identity reported significant negative relationships between Autistic identity and mental health outcomes, indicating that Autistic identity is a protective factor against poor mental health (Cage et al., 2018; K. Cooper et al., 2017; R. Cooper et al., 2021). This study suggests that Autistic identity as a protective factor may not extend to ED symptoms or may indicate that current scales that measure ED symptomology do not adequately capture ED symptoms in Autistic groups (e.g. rules around eating may be viewed as being part of being Autistic and thus might not be captured by a measure focusing on traditional ED symptom).

Sensory processing and ED symptomology

Sensory sensitivities and ED symptomology showed a positive relationship which supports previous findings about the role of sensory issues and eating behaviours in Autistic groups (Nimbley et al., 2022). It is worth noting, however, that although sensory sensitivity significantly explained variance in ED symptomology; when added to the regression model, it did not explain a large amount. It also did not significantly explain unique variance, indicating that although these concepts are related, sensory sensitivities do not appear to strongly predict ED symptomatology in autism.

The relationship between sensory sensitivities and AN has been highlighted (Brede et al., 2020); however, this has not been consistently corroborated (Kinnaird et al., 2020). In addition, there is even less research exploring the relationship between sensory sensitivities and other EDs, such as BED, and the few studies that have explored this have found no relationship (Kelly et al., 2013). Participants in this study reported either not having an ED or reported

a range of Eds, including binge disorder, BN, OSFED (other specified feeding and eating disorder), and also ARFID which is a Feeding Disorder (Hay et al., 2017). This may explain the weaker predictive value of sensory processing and may suggest that sensory issues are more prevalent in some EDs than others. Furthermore, this finding may also have been again caused by the EDE-QS measure being developed based on more typical ED presentations and may not adequately capture the range or differing ED presentations that are reported to co-occur with autism (Huke et al., 2013).

Strengths and limitations

It is essential to investigate the various factors that contribute to all types of EDs and disordered eating behaviours among Autistic people. This study has added to the research base by exploring sensory processing, Autistic identity and camouflaging factors for the first time in terms of ED symptomology. Despite this strength, there are limitations associated with the study that must be noted. This includes the limited availability of existing measures to capture Autistic identity. The SIS was not originally developed for measuring Autistic identity; however, it was the only appropriate tool available that had previously been adapted and used to measure Autistic identity (K. Cooper et al., 2017).

In addition, the EDE-QS which was used to capture ED symptomology contained items that aimed to measure constructs such as weight and shape concerns (Gideon et al., 2016). Therefore, it may not have been an appropriate measure for some of the eating symptomology experienced by the current Autistic participants, particularly if this ED symptomology is not driven by weight or shape concerns such as those who have ARFID (Hay et al., 2017). Recent research has improved understanding about the potential social and sensory motivations in Autistic people who have AN (Nimbley et al., 2023) and similar work needs to be carried out for other eating and/or feeding disorders.

The cross-sectional design and regression analyses method used in this study limits the conclusions that can be made in terms of the directionality of the relationships between camouflaging and sensory processing with ED symptomology. There may be other confounding variables that were not accounted for in this study that explain the relationships, for example, thwarted belongingness or cognitive factors.

This design of this study took participant burden into consideration and chose to use shortened versions of measures such as the SPQ-10 and the EDE-QS which displayed good reliability and validity (Prnjak et al., 2020). Despite a potential strength of this study being the online recruitment method (which allowed participants from various geographical locations and demographics to take part), this may have also been a limitation potentially causing a biased sample, with those most likely to have a strong Autistic identity to take part. Related to this a large demographic majority of this study were White (87.2%) and female (64.4%) which means that caution should be taken when applying the findings of these results to all Autistic people (those who have high support needs, trans and non-binary, children, etc.).

In addition to the potential bias in the current sample, no methods were used to verify clinical diagnoses of autism and self-identified Autistic people were also included. Participant recruitment routes were not recorded (due to the nature of the survey) and it is not known how many participants were recruited from the NHS or community. Also, despite the AQ-10 being used to indicate levels of autistic traits, it was not used as a screening tool as research has suggested its poor reliability in this regard (Taylor et al., 2020). Therefore, although this study endorsed a more inclusive research practice, caution must be warranted when interpreting the results.

Future research

Future research needs to begin looking at whether these factors or others (i.e. Autism acceptance or Autism collective self-esteem) predict or protect against ED symptomology and the different types of ED diagnoses. This study explored Autistic identity; however, future studies should also explore the potential negative impact that ED identity may have on Autistic people with a diagnosis of ED. As this study has been the first to demonstrate a relationship between camouflaging and ED in an Autistic population with and without diagnoses of ED, future studies could explore this relationship among Autistic people, including those with a clinical diagnosis of ED. Longitudinal research is required to determine the direction of the significant relationships found within this study. In addition to this, more underrepresented Autistic groups need to be included in ED research to better understand the underpinning mechanisms that contribute to ED diagnoses; these groups include those

who have non-White ethnicity, non-cis gender and those who have high support needs.

Recommendations

This study has highlighted the need for the development of reliable and validated measures to assess Autistic identity given importance of this concept in terms of SIT and autism being an historically stigmatised group. This study also emphasises the need for increased efforts and funding to reduce the stigma of autism and create a more neurodiversity-affirming environment. Reducing the stigma is likely to reduce the need for Autistic people to engage in camouflaging behaviours, leading to poor mental health outcomes. This is particularly important as this study adds to this literature by showing that camouflaging predicts ED symptomology. It may be worth clinicians working with Autistic people to consider screening for camouflaging behaviours to inform their assessments and formulation of mental health difficulties such as EDs. Finally, it is important to further develop and validate scales that measure ED symptomology such as the EDE-QS; to ensure that it measures the range of EDs with varying ED presentation in Autistic groups.

Conclusion

This study has uniquely added to understanding the relationship between autism and ED symptomatology by highlighting the impact that camouflaging behaviours and sensory processing have on ED symptomology in Autistic people. This emphasises the need to increase awareness of camouflaging behaviours and their impact on EDs symptoms in Autistic groups. Although Autistic identity was not identified as a significant factor in this study, more research should consider its measurement and the role it may play in different types of EDs. More quantitative and qualitative research should be conducted to extend the findings of this study and provide a more in-depth understanding of the roles of sensory processing and camouflaging in Autistic groups who have EDs.

Acknowledgements

The authors wish to thank the participants who gave up their time to take part in the current research.

Declaration of conflicting interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship and/or publication of this article: K.G.-S. and F.D. are supported by UK Research and Innovation

(MRC, ESRC, AHRC), the National Institute for Health and Care Research and the Medical Research Foundation as part of the EDAC network (grant number: MR/X03058X/1).

ORCID iDs

Tasha Suratwala https://orcid.org/0009-0007-4580-6744
Karri Gillespie-Smith https://orcid.org/0000-0002-3059-7632

Note

 The words Autism and Autistic have been capitalised throughout this article when referring to matters related to identity or Autistic community as this reflects the neurodivergent perspective that Autism is an identity rather than a medical condition.

References

- Adamson, J., Kinnaird, E., Glennon, D., Oakley, M., & Tchanturia, K. (2020). Carers' views on autism and eating disorders comorbidity: Qualitative study. *BJPsych Open*, 6(3), Article e51. https://doi.org/10.1192/bjo.2020.36
- Allison, C., Auyeung, B., & Baron-Cohen, S. (2012). Toward brief 'red flags' for autism screening: The short autism spectrum quotient and the short quantitative checklist in 1,000 cases and 3,000 controls. *Journal of the American Academy of Child and Adolescent Psychiatry*, 51, 202–212.e7. https://doi.org/10.1016/j.jaac.2011.11.003
- American Psychiatric Association. (2013). *Diagnostic and statistical manual of mental disorders* (5th ed.).
- Bargiela, S., Steward, R., & Mandy, W. (2016). The experiences of late-diagnosed women with autism spectrum conditions: An investigation of the female autism phenotype. *Journal of Autism and Developmental Disorders*, 46, 3281–3294. https://doi.org/10.1007/s10803-016-2872-8
- Barnett, A., Edwards, K., Harper, R., Evans, E., Alexander, D., Choudhary, M., & Richards, G. (2021). The association between autistic traits and disordered eating is moderated by sex/gender and independent of anxiety and depression. *Journal of Autism and Developmental Disorders*, *51*(6), 1866–1879. https://doi.org/10.1007/s10803-020-04669-z
- Beck, J. S., Lundwall, R. A., Gabrielsen, T., Cox, J. C., & South, M. (2020). Looking good but feeling bad: 'camouflaging' behaviors and mental health in women with autistic traits. *Autism*, 24(4), 809–821. https://doi.org/10.1177/ 1362361320912147
- Bentz, M., Jepsen, J. R. M., Pedersen, T., Bulik, C. M., Pedersen, L., Pagsberg, A. K., & Plessen, K. J. (2017). Impairment of social function in young females with recent-onset anorexia nervosa and recovered individuals. *Journal of Adolescent Health*, 60(1), 23–32. https://doi.org/10.1016/j.jadohealth. 2016.08.011
- Botha, M., & Frost, D. M. (2020). Extending the minority stress model to understand mental health problems experienced by the autistic population. *Society and Mental Health*, *10*(1), 20–34. https://doi.org/10.1177/2156869318804297
- Brand-Gothelf, A., Parush, S., Eitan, Y., Admoni, S., Gur, E., & Stein, D. (2016). Sensory modulation disorder symptoms in anorexia nervosa and bulimia nervosa: A pilot study. International. *Journal of Eating Disorders*, 49(1), 59–68. https://doi.org/10.1002/eat.22460

- Brede, J., Babb, C., Jones, C., Elliott, M., Zanker, C., Tchanturia, K., & Mandy, W. (2020). 'For me, the anorexia is just a symptom, and the cause is the autism': Investigating restrictive eating disorders in autistic women. *Journal of Autism and Developmental Disorders*, 50(12), 4280–4296. https://doi.org/10.1007/s10803-020-04479-3
- Cage, E., Di Monaco, J., & Newell, V. (2018). Experiences of autism acceptance and mental health in autistic adults. *Journal of Autism and Developmental Disorders*, 48(2), 473–484. https://doi.org/10.1007/s10803-017-3342-7
- Carter Leno, V., Micali, N., Bryant-Waugh, R., & Herle, M. (2022). Associations between childhood autistic traits and adolescent eating disorder behaviours are partially mediated by fussy eating. *European Eating Disorders Review*, 30(5), 604–615.
- Christensen, S. S., Bentz, M., Clemmensen, L., Strandberg-Larsen, K., & Olsen, E. M. (2019). Disordered eating behaviours and autistic traits—Are there any associations in nonclinical populations? A systematic review. *European Eating Disorders Review*, 27(1), 8–23. https://doi.org/10.1002/erv.2627
- Cohen, J. (1988). Statistical power analysis for the behavioural sciences (2nd ed.). L. Erlbaum Associates.
- Cook, J., Hull, L., Crane, L., & Mandy, W. (2021). Camouflaging in autism: A systematic review. *Clinical Psychology Review*, 89, Article 102080. https://doi.org/10.1016/j.cpr.2021.102080
- Cooper, K., Smith, L. G. E., & Russell, A. (2017). Social identity, self-esteem, and mental health in autism. *European Journal of Social Psychology*, 47(7), 844–854. https://doi.org/10.1002/ejsp.2297
- Cooper, R., Cooper, K., Russell, A. J., & Smith, L. G. E. (2021). 'I'm proud to be a little bit different': The effects of autistic individuals' perceptions of autism and autism social identity on their collective self-esteem. *Journal of Autism and Developmental Disorders*, 51(2), 704–714. https://doi.org/10.1007/s10803-020-04575-4
- Crane, L., Goddard, L., & Pring, L. (2009). Sensory processing in adults with autism spectrum disorders. *Autism*, 13(3), 215–228. https://doi.org/10.1177/1362361309103794
- Dandil, Y., Baillie, C., & Tchanturia, K. (2019). Cognitive remediation therapy as a feasible treatment for a young person with anorexia nervosa and autism spectrum disorder comorbidity: A case study. Clinical Case Studies, 19(2), 115–132. https://doi.org/10.1177/1534650119890425
- Faul, F., Erdfelder, E., Buchner, A., & Lang, A. G. (2009). Statistical power analyses using G* Power 3.1: Tests for correlation and regression analyses. *Behavior Research Methods*, *41*(4), 1149–1160. https://link.springer.com/article/10.3758/BRM.41.4.1149
- Fletcher-Watson, S., Adams, J., Brook, K., Charman, T., Crane, L., Cusack, J., . . . Pellicano, E. (2019). Making the future together: Shaping autism research through meaningful participation. *Autism*, *23*(4), 943–953. https://doi.org/10.1177/1362361318786721
- Gideon, N., Hawkes, N., Mond, J., Saunders, R., Tchanturia, K., & Serpell, L. (2016). Development and psychometric validation of the EDE-QS, a 12 item short form of the Eating Disorder Examination Questionnaire (EDE-Q). *PLOS ONE*, 11(5), Article e0152744. https://doi.org/10.1371/journal. pone.0152744

Green, S. B. (1991). How many subjects does it take to do a regression analysis. *Multivariate Behavioral Research*, 26(3), 499–510. https://doi.org/10.1207/s15327906mbr2603_7

- Greenberg, D. M., Warrier, V., Allison, C., & Baron-Cohen, S. (2018). Testing the empathizing–Systemizing theory of sex differences and the extreme male brain theory of autism in half a million people. *Proceedings of the National Academy of Sciences*, 115(48), 12152–12157. https://doi.org/10.1073/pnas.1811032115
- Hay, P., Mitchison, D., Collado, A. E. L., González-Chica, D. A., Stocks, N., & Touyz, S. (2017). Burden and health-related quality of life of eating disorders, including Avoidant/ Restrictive Food Intake Disorder (ARFID), in the Australian population. *Journal of Eating Disorders*, 5(1), 1–10. https:// doi.org/10.1186/s40337-017-0149-z
- Hendrickx, S. (2015). Women and girls with autism spectrum disorder: Understanding life experiences from early childhood to old age. Jessica Kingsley Publishers.
- Hiller, R., & Pellicano, L. (2013). Anorexia and autism: A cautionary note. *The Psychologist*, 26(11), 780. https://www.scopus.com/record/display.uri?eid=2-s2.0-84894495486&origin=inward&txGid=907396838b8bbd05e1b257d57ada0699
- Huke, V., Turk, J., Saeidi, S., Kent, A., & Morgan, J. F. (2013). Autism spectrum disorders in eating disorder populations: A systematic review. *European Eating Disorders Review*, 21(5), 345–351. https://doi.org/10.1002/erv.2244
- Hull, L., & Mandy, W. (2017). Protective effect or missed diagnosis? Females with autism spectrum disorder. Future Neurology, 12(3), 159–169. https://doi.org/10.2217/fnl-2017-0006
- Hull, L., Mandy, W., Lai, M. C., Baron-Cohen, S., Allison, C., Smith, P., & Petrides, K. V. (2019). Development and validation of the Camouflaging Autistic Traits Questionnaire (CAT-Q). *Journal of Autism and Developmental Disorders*, 49(3), 819–833. https://doi.org/10.1007/s10803-018-3792-6
- Kelly, N. R., Bulik, C. M., & Mazzeo, S. E. (2013). Executive functioning and behavioral impulsivity of young women who binge eat. *The International Journal of Eating Disorders*, 46(2), 127–139. https://doi.org/10.1002/eat.22096
- Kerr-Gaffney, J., Halls, D., Harrison, A., & Tchanturia, K. (2020). Exploring relationships between autism spectrum disorder symptoms and eating disorder symptoms in adults with anorexia nervosa: A network approach. Frontiers in Psychiatry, 11, Article 401. https://doi.org/10.3389/fpsyt.2020.00401
- Kinnaird, E., Dandil, Y., Li, Z., Smith, K., Pimblett, C., Agbalaya, R., . . . Tchanturia, K. (2020). Pragmatic sensory screening in anorexia nervosa and associations with autistic traits. *Journal of Clinical Medicine*, 9(4), Article 1182. https://doi. org/10.3390/jcm9041182
- Kinnaird, E., Norton, C., Stewart, C., & Tchanturia, K. (2019). Same behaviours, different reasons: What do patients with co-occurring anorexia and autism want from treatment? *International Review of Psychiatry*, 31(4), 308–317. https://doi.org/10.1080/09540261.2018.1531831
- Kinnaird, E., Norton, C., & Tchanturia, K. (2017). Clinicians' views on working with anorexia nervosa and autism spectrum disorder comorbidity: A qualitative study. BMC Psychiatry, 17(1), Article 292. https://doi.org/10.1186/s12888-017-1455-3
- Lai, M. C., Kassee, C., Besney, R., Bonato, S., Hull, L., Mandy, W., . . . Ameis, S. H. (2019). Prevalence of co-occurring

- mental health diagnoses in the autism population: A systematic review and meta-analysis. *Lancet Psychiatry*, 6(10), 819–829. https://doi.org/10.1016/S2215-0366(19)30289-5
- Lai, M. C., Lombardo, M. V., Ruigrok, A. N., Chakrabarti, B., Auyeung, B., & Szatmari, A. I. M. S., . . . Consortium, . (2017). Quantifying and exploring camouflaging in men and women with autism. *Autism*, 21(6), 690–702. https:// dx.doi.org/10.1177/1362361316671012
- Lavender, J. M., Brown, T. A., & Murray, S. B. (2017). Men, muscles, and eating disorders: An overview of traditional and muscularity-oriented disordered eating. *Current Psychiatry Reports*, 19(6), Article 32. https://doi.org/10.1007/s11920-017-0787-5
- Leach, C. W., Van Zomeren, M., Zebel, S., Vliek, M. L., Pennekamp, S. F., Doosje, B., . . . Spears, R. (2008). Grouplevel self-definition and self-investment: A hierarchical (multicomponent) model of in-group identification. *Journal* of *Personality and Social Psychology*, 95(1), Article 144. https://doi.org/10.1037/0022-3514.95.1.144
- MacLeod, A., Lewis, A., & Robertson, C. (2013). 'Why should I be like bloody Rain Man?' Navigating the autistic identity. *British Journal of Special Education*, 40(1), 41–49. https://doi.org/10.1111/1467-8578.12015
- Manasse, S. M., Forman, E. M., Ruocco, A. C., Butryn, M. L., Juarascio, A. S., & Fitzpatrick, K. K. (2015). Do executive functioning deficits underpin binge eating disorder? A comparison of overweight women with and without binge eating pathology. *International Journal of Eating Disorders*, 48(6), 677–683. https://doi.org/10.1002/eat.22383
- McDonald, T. A. M. (2020). Autism identity and the 'lost generation': Structural validation of the Autism Spectrum Identity Scale and comparison of diagnosed and self-diagnosed adults on the autism spectrum. *Autism in Adulthood*, *2*(1), 13–23.
- Milner, V., McIntosh, H., Colvert, E., & Happe, F. (2019). A qualitative exploration of the female experience of autism spectrum disorder (ASD). *Journal of Autism and Developmental Disorders*, 49(6), 2389–2402. https://doi.org/10.1007/s10803-019-03906-4
- Murray, S. B., Nagata, J. M., Griffiths, S., Calzo, J. P., Brown, T. A., Mitchison, D., . . . Mond, J. M. (2017). The enigma of male eating disorders: A critical review and synthesis. *Clinical Psychology Review*, *57*, 1–11. https://doi.org/10.1016/j.cpr.2017.08.001
- Nazar, B. P., Peynenburg, V., Rhind, C., Hibbs, R., Schmidt, U., Gowers, S., . . . Treasure, J. (2018). An examination of the clinical outcomes of adolescents and young adults with broad autism spectrum traits and autism spectrum disorder and anorexia nervosa: A multi centre study. *International Journal of Eating Disorders*, 51(2), 174–179. https://doi.org/10.1002/eat.22823
- NICE (2017) Eating disorders: recognition and treatment.

 National Institute for Health and Care Excellence. http://www.nice.org.uk
- Nickel, K., Maier, S., Endres, D., Joos, A., Maier, V., Tebartz van Elst, L., & Zeeck, A. (2019). Systematic review: Overlap between eating, autism spectrum, and attention-deficit/ hyperactivity disorder. *Frontiers in Psychiatry*, 10, Article 708.https://doi.org/10.3389/fpsyt.2019.00708
- Nielsen, S., Anckarsäter, H., Gillberg, C., Gillberg, C., Råstam, M., & Wentz, E. (2015). Effects of autism spectrum disorders on outcome in teenage-onset anorexia nervosa evaluated by

the Morgan-Russell outcome assessment schedule: A controlled community-based study. *Molecular Autism*, *6*, 1–10. https://doi.org/10.1186/s13229-015-0013-4

- Nielsen, S., Dobrescu, S. R., Dinkler, L., Gillberg, C., Gillberg, C., Råstam, M., & Wentz, E. (2022). Effects of autism on 30-year outcome of anorexia nervosa. *Journal of Eating Disorders*, 10(1), 1–14. https://doi.org/10.1186/s40337-021-00518-1
- Nimbley, E., Gillespie-Smith, K., Duffy, F., Maloney, E., Ballantyne, C., & Sharpe, H. (2023). 'It's not about wanting to be thin or look small, it's about the way it feels': An IPA analysis of social and sensory differences in autistic and non-autistic individuals with anorexia and their parents. *Journal of Eating Disorders*, 11(1), Article 89. https://doi.org/10.1186/s40337-023-00813-z
- Nimbley, E., Golds, L., Sharpe, H., Gillespie-Smith, K., & Duffy, F. (2022). Sensory processing and eating behaviours in autism: A systematic review. *European Eating Disorders Review*, *30*(5), 538–559. https://doi.org/10.1002/erv.2920
- Overton, G. L., Marsa-Sambola, F., Martin, R., & Cavenagh, P. (2023). Understanding the self-identification of autism in adults: A scoping review. *Review Journal of Autism and Developmental Disorders*, 1–21. https://doi.org/10.1007/s40489-023-00361-x
- Perry, E., Mandy, W., Hull, L., & Cage, E. (2022). Understanding camouflaging as a response to autism-related stigma: A social identity theory approach. *Journal of Autism and Developmental Disorders*, 52(2), 800–810. https://doi.org/10.1007/s10803-021-04987-w
- Prnjak, K., Mitchison, D., Griffiths, S., Mond, J., Gideon, N., Serpell, L., & Hay, P. (2020). Further development of the 12-item EDE-QS: Identifying a cut-off for screening purposes. *BMC Psychiatry*, 20(1), 1–7. https://doi.org/10.1186/ s12888-020-02565-5
- Sedgewick, F., Leppanen, J., & Tchanturia, K. (2021). Gender differences in mental health prevalence in autism. *Advances* in *Autism*, 7(3), 208–224. https://kclpure.kcl.ac.uk/ws/portalfiles/portal/126997323/AS MH rates Final.pdf

- Solmi, F., Bentivegna, F., Bould, H., Mandy, W., Kothari, R., Rai, D., . . . Lewis, G. (2021). Trajectories of autistic social traits in childhood and adolescence and disordered eating behaviours at age 14 years: A UK general population cohort study. *Journal of Child Psychology and Psychiatry*, 62(1), 75–85. https://doi.org/10.1111/jcpp.13255
- Striegel-Moore, R. H., & Bulik, C. M. (2007). Risk factors for eating disorders. *American Psychologist*, 62(3), 181–198. https://psycnet.apa.org/fulltext/2007-04834-005.pdf
- Tajfel, H., Turner, J. C., Austin, W. G., & Worchel, S. (1979).
 An integrative theory of intergroup conflict. In H. Tajfel & J. C. Turner (Eds.), *Organizational identity: A reader* (pp. 56–65). Oxford University Press.
- Taylor, E. C., Livingston, L. A., Clutterbuck, R. A., Shah, P., & Payne, C. (2020). Psychometric concerns with the 10-item Autism-Spectrum Quotient (AQ10) as a measure of trait autism in the general population. *Experimental Results*, *1*, Article e3. https://doi.org/10.1017/exp.2019.3
- Tchanturia, K., Dandil, Y., Li, Z., Smith, K., Leslie, M., & Byford, S. (2021). A novel approach for autism spectrum condition patients with eating disorders: Analysis of treatment cost-savings. *European Eating Disorders Review*, 29(3), 514–518. https://doi.org/10.1002/erv.2760
- Westwood, H., Mandy, W., Simic, M., & Tchanturia, K. (2018).
 Assessing ASD in adolescent females with anorexia nervosa using clinical and developmental measures: A preliminary investigation. *Journal of Abnormal Child Psychology*, 46, 183–192. https://link.springer.com/article/10.1007/s10802-017-0301-x
- Westwood, H., Mandy, W., & Tchanturia, K. (2017). Clinical evaluation of autistic symptoms in women with anorexia nervosa. *Molecular Autism*, 8, 1–9. https://doi.org/10.1186/ s13229-017-0128-x
- Zeidan, J., Fombonne, E., Scorah, J., Ibrahim, A., Durkin, M. S., Saxena, S., . . . Elsabbagh, M. (2022). Global prevalence of autism: A systematic review update. *Autism Research*, 15(5), 778–790. https://doi.org/10.1002/aur.2696